Home environmental assessments and modification delivered by occupational therapists to reduce falls in people aged 65 years and over: the OTIS RCT

Sarah Cockayne,^{1*} Alison Pighills,^{2,3} Joy Adamson,¹ Caroline Fairhurst,¹ Shelley Crossland,⁴ Avril Drummond,⁵ Catherine E Hewitt,¹ Sara Rodgers,¹ Sarah J Ronaldson,¹ Jennifer McCaffery,¹ Katie Whiteside,¹ Arabella Scantlebury,¹ Lyn Robinson-Smith,¹ Ann Cochrane,¹ Sarah E Lamb,⁶ Sophie Boyes,⁷ Simon Gilbody,¹ Clare Relton⁸ and David J Torgerson¹ on behalf of the OTIS study

¹York Trials Unit, Department of Health Sciences, University of York, York, UK ²Mackay Institute of Research and Innovation, Queensland Health, Mackay Base Hospital, Mackay, QLD, Australia ³Division of Tropical Health and Medicine, James Cook University, Townsville,

- QLD, Australia
- ⁴Leicestershire Partnership NHS Trust, Leicester, UK
- ⁵School of Health Sciences, University of Nottingham, Nottingham, UK
- ⁶Institute of Health Research, College of Medicine and Health, University of Exeter, Exeter, UK
- ⁷York Teaching Hospital NHS Foundation Trust, York, UK
- ⁸Institute of Population Health Sciences, Queen Mary University of London, London, UK

*Corresponding author sarah.cockayne@york.ac.uk

Declared competing interests of authors: Catherine E Hewitt declares membership of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Commissioning Sub-board (EOI) (2016–17). Sarah E Lamb was a member of Clinical Trial Units funded by NIHR and declares previous membership of the HTA Additional Capacity Funding Board (2012–15), the HTA End of Life Care and Add-on Studies Board (2015), the HTA Prioritisation Group Board and the HTA Trauma Board. Simon Gilbody declares that he was deputy chairperson of the NIHR HTA programme Commissioning Committee from 1 November 2016 to 30 October 2019. Joy Adamson is a member of HTA Commissioning Board (August 2017–present). David J Torgerson declares that he was a member of Clinical Trial Units funded by NIHR, a member of various HTA commissioning boards and a member of the NIHR Clinical Trial Units Standing Advisory Committee (2010–14). Avril Drummond declares that she is a member (June 2017–present) of the NIHR Academy, Clinical Lectureship and Senior Lectureship Panel and is now deputy chairperson (January 2020–present).

Published July 2021 DOI: 10.3310/hta25460

Plain English summary

The OTIS RCT Health Technology Assessment 2021; Vol. 25: No. 46 DOI: 10.3310/hta25460

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

F alling is a common problem among older people. In fact, up to half of people aged over 80 years fall each year, with most falls happening inside the home. Unfortunately, some falls cause serious injuries, such as broken bones. People often think that falls are part of getting older and that little can be done to stop them from falling. However, there are many ways to reduce falls. The Occupational Therapist Intervention Study (OTIS) explored whether or not a home assessment visit by an NHS occupational therapist could reduce falls among older people who are likely to be at risk of falling.

In total, 1331 people aged 65 years or older living in England took part in the study. These people were all sent an Age UK leaflet about how to prevent falls, and 430 people were selected at random to receive a visit from an occupational therapist. The occupational therapist assessed their homes for hazards, such as slippery floors or poor lighting, and made suggestions for changes. We collected information from participants using monthly falls calendars and postal questionnaires to ask them about their falls, their quality of life, how often they used NHS services and how often they used paid care workers. We also asked them about whether they had had equipment and adaptations installed as a result of the assessments.

We found that the home assessment visits did not reduce the number of falls people had or make any difference to participants' quality of life. However, many of the recommendations made by the occupational therapists were not carried out. The home assessment visits by an occupational therapist were not good value for money.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Cockayne *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/49/149. The contractual start date was in June 2016. The draft report began editorial review in January 2020 and was accepted for publication in July 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Cockayne *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk